2471

403, 2022, 4, Down



J

GeoHealth

RESEARCH ARTICLE

10.1029/2022GH000611

Key Points:

- The long-term trend of dry spell length (DSL) during 1976–2019 was examined using the global observation data set of daily precipitation
- The global mean DSL significantly increased during 1976–1998 whereas that showed no significant trend during 1999–2019
- The variation of annual precipitation can modulate the spatiotemporal variations of DSL

Correspondence to:

H. Lu, luhaibo@mail.sysu.edu.cn

Citation:

Wang, X., Lu, H., & Yuan, W. (2022). Inter-annual variations of precipitation modulate the dry spell length. *GeoHealth*, *6*, e2022GH000611. https://doi. org/10.1029/2022GH000611

Received 26 FEB 2022 Accepted 29 MAR 2022

Author Contributions:

Conceptualization: Haibo Lu, Wenping Yuan

Data curation: Xiaoyuan Wang Formal analysis: Xiaoyuan Wang, Haibo Lu

Methodology: Haibo Lu, Wenping Yuan Project Administration: Wenping Yuan Supervision: Wenping Yuan Validation: Xiaoyuan Wang Writing – original draft: Haibo Lu Writing – review & editing: Haibo Lu, Wenping Yuan

© 2022 The Authors. GeoHealth published by Wiley Periodicals LLC on behalf of American Geophysical Union. This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Inter-Annual Variations of Precipitation Modulate the Dry Spell Length

Xiaoyuan Wang¹, Haibo Lu^{2,3}, and Wenping Yuan³

¹State Key Laboratory of Earth Surface Processes and Resource Ecology, Faculty of Geographical Science, Beijing Normal University, Beijing, China, ²Zhuhai Branch of State Key Laboratory of Earth Surface Processes and Resource Ecology, Department of Geography, Faculty of Arts and Sciences, Beijing Normal University, Zhuhai, China, ³Southern Marine Science and Engineering Guangdong Laboratory (Zhuhai), School of Atmospheric Sciences, Sun Yat-sen University, Zhuhai, China

Abstract Dry spell length (DSL), consecutive non-rainy days between two precipitation events, play an important role in regulating soil moisture dynamics, terrestrial energy exchange as well as vegetation growth. According to the Clausius-Clapeyron (C-C) relationship, global warming can result in prolonged DSL. However, usually the amount of precipitation and its characteristics coincidentally varied with the changes of DSL under global warming, it remains unclear how the inter-annual variation of precipitation interacts with the evolution of dry spells. In this study, the global long-term in-situ observation data set of daily precipitation during 1976–2019 was used to examine the spatiotemporal trends of growing season DSL and precipitation. Our results showed that the global mean growing season DSL significantly increased by 0.3 days decade⁻¹ during 1976–1998 while no significant trend of that was observed during 1999–2019. In contrast, the growing season precipitation (Prec_GS) showed no significant trend in 1976-1998 whereas significant increase trend of that was observed in 1999–2019. To explore the impacts of precipitation on the evolution of dry spells, we examined the relationship between the growing season DSL and Prec_GS. We found that prevalent negative relationship was observed between growing season DSL and Prec_GS in 88% and 86% stations during the period of 1976-1998 and 1999-2019, respectively. Spatially, the mean annual Prec_GS and DSL showed significantly negative relationship, that is, the stations with more precipitation showed shorter DSL in growing season, and vice versa. The changes of mean annual Prec GS explained 81% spatial variation of growing season DSL. Moreover, during the period of 1999-2019 significant increase of precipitation frequency and decrease of dry day frequency were also observed in addition to the increase of Prec_GS in this period. The decreased dry day frequency further resulted in the decrease of growing season DSL. By excluded the impacts of precipitation, the DSL/Prec_GS ratio showed significant decreasing trend during 1999–2019. Our study suggested that the spatiotemporal variations of DSL were modulated by the variation of precipitation. The impacts of precipitation changes on ecosystem by altering the dry spell evolution should be considered in modeling the terrestrial carbon and hydrological cycling in response to climate changes.

Plain Language Summary Dry spell length (DSL) indicates the duration between two precipitation events. As an important precipitation characteristic, DSL can regulate terrestrial carbon-water cycling. However, it is still unclear how climate change influenced the evolution of DSL. Here, we examined the spatiotemporal trends of DSL globally during 1976–2019 using the long-term in-situ observation data set of daily precipitation. We found that the global mean growing season DSL significantly increased during 1976–1998 whereas no significant trend was observed during 1999–2019. The significant increase in the amount of precipitation during 1999–2019 contribute to the diminished increasing trend of DSL in this period. Our study suggested that the spatiotemporal variations of DSL were modulated by the variation of precipitation. The impacts of precipitation changes on ecosystem by altering the dry spell evolution should be taken into account in modeling the terrestrial carbon and hydrological cycling in response to climate changes.

1. Introduction

Climate change profoundly alters global hydrologic cycling (Lambert et al., 2008; Loaiciga et al., 1996; Wentz et al., 2007). Dry spell length (DSL), consecutive non-rainy days between two precipitation events, which associates with precipitation frequency, directly influence soil moisture dynamics, terrestrial energy exchange as well as vegetation growth (Chou et al., 2012). Theoretically, global warming can increase atmospheric water

vapor holding capacity by 7%/K according to the Clausius-Clapeyron (C-C) relationship (Trenberth et al., 2005). However, global evapotranspiration tends to increase at a lower rate than the C-C relationship restricted by global energy budget and the slower ocean warming (Trenberth et al., 2003). Given that, it would take a relative long recharge time for water vapor to replenish the atmosphere, that results in long dry spells. The extended dry spell could trigger drought, heatwaves and further impact on terrestrial carbon uptake (Ye & Fetzer, 2019). Thus, quantifying the spatiotemporal variation of DSL is essential for revealing the impacts of climate change on terrestrial carbon and water cycling.

The extend DSL across different geographical regions has been reported by previous studies basing on ground observations and global climate model experiments. Giorgi et al. (2011) indicated an overall increasing trend of either DSL or precipitation intensity during 1970–2000 across all regions except Australia and South America, using global ground station and gridded precipitation observation datasets. In North America continent, by analyzing the daily precipitation records during 1967–2006 from ground observing stations, it was found that the mean duration of consecutive dry episodes significantly increased in eastern and southwestern United States in recent decades (Groisman & Knight, 2008). In Europe, Zolina et al. (2013) found that DSL was extended by 1.5 to more than 2.5 days in central western and northern Europe during the past 60 years by analyzing data from 699 European rain gauge stations. Moreover, changes of DSL in different regions were also predicted by global and regional atmospheric models. Climate models indicate that longer duration of DSL are expected in Canada and tropical Africa (Bouagila & Sushama, 2013; Sushama et al., 2010). Based on the regional climate models, the duration of dry spells in summer was projected to increase by the end of this century in Switzerland (Fischer et al., 2015).

However, it remains unclear how the variations in amount of precipitation interact with DSL. Generally, the spatiotemporal changes of DSL were accompanied by the variations of precipitation in a lot of regions. Climate change alters both spatial and temporal variations of precipitation (Dore, 2005; Fischer et al., 2015). The inter-annual variations of precipitation are expected to alter the frequency of precipitation and dry days, that would further impact on the evolution of dry spells. For example, in central and northern Europe, the length of dry spells increased in recent decades, at the same time an increasing trend of precipitation (more than 20 mm per decade) has been observed (Caloiero et al., 2018; Zolina et al., 2013). The prolonged summer dry spell duration in last decades also have been observed in Russia (Ye & Fetzer, 2019), which was associated with decreased summer precipitation frequency and increased intensity during the same period (Lebedeva et al., 2016; Ye et al., 2016). Hence, it is necessary to examine the relationship between the patterns of annual precipitation and the evolution of dry spells.

In this study, the global long-term observations data set of daily precipitation deriving from in-situ weather stations was used to retrieve growing season DSL and precipitation characteristics. The primary objectives of this study were to (a) estimate the global spatiotemporal patterns of DSL and precipitation in growing season and (b) examine the relationship between the patterns of growing season precipitation and the evolution of dry spells.

2. Data and Methods

2.1. Global Precipitation Observation Data Set

In this study, the daily precipitation observations were used to explore the spatiotemporal changes of DSL and precipitation, the database was derived from the Global Surface Summary of Day (GSOD) data set (https://www.ncei.noaa.gov/data/global-summary-of-the-day/). The GSOD data set complied precipitation records covering more than 9,000 stations across the globe, from 1929 to the present, with data from 1973 to the present being the most complete. To extract intact DSL information, we selected the precipitation observations from the GSOD data set by the following criteria: (a) the daily precipitation observation records of each year were complete (i.e., 365 or 366 records each year in a station) and (b) each station which was selected should have at least 35 years complete precipitation observations; In total, 1,596 stations were selected in this analysis, ranging from 1976 to 2019.

The DSL during growing season was extracted for each observation station. We use a threshold of 0.1 mm day^{-1} to define a precipitation event (Sun et al., 2007), the day with daily precipitation less than 0.1 mm is defined as a dry day (non-rainy day). The number of consecutive non-rainy days between two precipitation events which was defined as DSL was extracted. Basing on that, we derived the mean DSL of growing season. The precipitation



frequency and dry day frequency during growing season were calculated by the ratio of rainy day and dry day to growing season days, respectively. In this study, the growing season is composed of all months where the mean monthly air temperature was above 0° C (Lu et al., 2021). The precipitation amount during growing season (Prec_GS) was also estimated for each station.

2.2. Data Analysis

In this study, we deployed the piecewise linear regression approach to detect the changes in trends of DSL, Prec_GS, precipitation frequency and dry day frequency in growing season during 1976–2019 (Toms & Lesperance, 2003; Yuan et al., 2019). The tipping point year were detected using following equations:

$$y = \begin{cases} \beta_0 + \beta_1 t + \varepsilon, & t \le \alpha \\ \beta_0 + \beta_1 t + \beta_2 (t - \alpha) + \varepsilon, & t > \alpha \end{cases}$$
(1)

where y indicates variables of the DSL, Prec_GS, precipitation frequency and dry day frequency in growing season; t is the year; α indicates the estimated the tipping point year; β_0 , β_1 , and β_2 are the regression coefficients; ε is the residual of the linear regression. β_1 and $\beta_1 + \beta_2$ represents the linear trend of each investigated variable before and after the tipping point year, respectively. α and other regression coefficients were estimated by least squares linear regression. The 5-year running mean were used 0.05 to quantify the tipping point year α . The statistical significance leval is at $P \le 0.05$.

The linear regression was performed to examine the temporal trend of DSL, Prec_GS, precipitation frequency and dry day frequency in growing season during the period of 1976–1998 and 1999–2019. For each station, only when there were at least 10 years valid data during the period (1976–1998 and 1999–2019) would the regression analysis be performed.

To examine the association between DSL with changes of precipitation during growing season, the linear and nonlinear regression analysis also was applied to explore the relationship between Prec_GS and DSL, Prec_GS and dry day frequency and dry day frequency and DSL, respectively.

3. Results

3.1. Changes of Global DSL and Precipitation Characteristics

The global mean trends of DSL, precipitation, precipitation frequency and dry day frequency were estimated from the precipitation observations of 1,596 stations. The observed global mean of growing season DSL showed significant increase by 0.3 days decade⁻¹ during 1976–1998 while no significant trend of that was shown during 1999–2019 (Figure 1a). On the contrary, there was no significant trend of growing season precipitation was observed during 1976–1998, whereas significant increase of that showed in 1999–2019 (Figure 1b). In line with the trend of precipitation, precipitation frequency and dry day frequency significantly increased and decreased during the period of 1999–2019, respectively (Figures 1c and 1d).

Spatially, the increasing trends of DSL in growing season were observed over 54% stations during 1976–1998, whereas only 37% stations showed increasing trend during 1999–2019. The diminished extending trends of DSL during the latter period mostly were observed in Europe (Figures 2a and 2b). Precipitation in growing season showed more obviously increasing trends during 1999–2019 (75% stations) than that in 1999–2019 (54% stations; Figures 2c and 2d). The increasing growing season precipitation mainly located in Europe and northwestern China, accordingly, decreasing trends of dry day frequency were observed in those regions (Figures 2e and 2f).

3.2. Associations of DSL With Changing Precipitation

To explore the impact of changes in growing season precipitation and dry day frequency on DSL, the linear regression was performed for each observation station. Over 88% stations showed negative relationship between DSL and growing season precipitation during study periods (Figure 3a). 54% stations which showed negative relationship between the dry day frequency and growing season precipitation during the entire study period





Figure 1. (a) Observed global mean dry spell length anomaly, (b) precipitation amount (Prec_GS) anomaly, (c) precipitation frequency, and (d) dry day frequency trends in growing season during 1976–2019.

(1976–2019). Nevertheless, the percentage of stations which showed significant (P < 0.05) negative relationship increased from 5% during 1976–1998 to 30% during 1999–2019, indicating that the increased growing season precipitation during the latter period resulted in less dry day frequency in more stations (Figure 3b). Comparing with the period of 1976–1998, more stations showed positive relationship of DSL and dry day frequency, especially the of percentage of stations which showed significant negative relationship sharply decreased from 21% during 1976–1998 to 6% during 1999–2019 (Figure 3c), suggesting that the decreased dry day frequency further induced shorter DSL. Moreover, spatially, significant negative relationship between averaged DSL and precipitation gradients was observed. The averaged DSL significantly decreased with the increase of mean growing





Figure 2. (a and b) Observed temporal trends of dry spell length, (c and d) precipitation, and (e and f) dry day frequency in growing season during the period of (left) 1976–1998 and (right) 1999–2019. The small insets indicate the percentage of significant increase (P < 0.05), non-significant increase, significant (P < 0.05), decrease and non-significant decrease stations.

season precipitation (Figure 4a). Changes of mean growing season precipitation explained 81% spatial variation of DSL, suggesting that growing season precipitation is one of the important regulators on DSL. Averaged DSL also showed significant positive correlation with dry day frequency spatially (Figure 4b).

To exclude the impact of growing season precipitation on DSL, we examined the temporal trends of the ratio of DSL to growing season precipitation (DSL/Prec_GS) for each station during the two periods. Our results indicated there was no significant difference between the trends of DSL and DSL/Prec_GS ratio during 1976–1998, 55% stations showed increasing trends in DSL/Prec_GS, which was very close to the percentage of 57% for DSL trend during this period (Figure 5a). Whereas, in the period of 1999–2019 where the growing season precipitation significantly increased, only 37% stations showed an increasing trend in DSL/Prec_GS ratio, this percentage is 18% less than that for DSL trend (increasing trend over 55% stations; Figure 5b). By excluded the impact of precipitation, the global mean DSL/Prec_GS ratio showed a significantly decreasing trend during 1999–2019, whereas during this period the global mean DSL had no significant trend (Figure 5c). The results suggested that the increase of growing season precipitation can result in shorter DSL during the period of 1999-2019. Another line evidence also supported this conclusion. To validate the regulation of growing season precipitation on DSL, stations which experienced significant (P < 0.1) changes in growing season precipitation were selected to examine the trend of DSL in each station. If the variations of growing season precipitation can exert negative effects on DSL, the DSL trends should be negative with the trend of growing season precipitation in those selected stations. The negative relationship between DSL trends and growing season precipitation trends was observed in 78% and 73% stations during the period of 1976–1998 and 1999–2019, respectively (Figure 6). In other words, the significant changes of growing season precipitation regulated the variation of DSL (i.e., significant increase of precipitation results in decrease of DSL, and vice versa) in most of the stations.





Figure 3. (a) The dependency of dry spell length (DSL) on precipitation, (b) dry day frequency on precipitation, and (c) DSL on dry day frequency in growing season during 1976–2019 over all stations. The probability density functions in left panels show the distribution of coefficients of linear regression between (a) DSL and precipitation, (b) dry day frequency and precipitation, and (c) DSL and dry day frequency, respectively. The stacked bar chart in right panels indicate the statistical results of the coefficients at significant level of 0.05 and 0.1.





Figure 4. (a) The spatial dependency of dry spell length (DSL) on precipitation and (b) dry day frequency in growing season over all stations, respectively. The gray dots in (a) indicate the averaged DSL over all stations within each precipitation bin, the bin size is 100 mm. The gray dots in (b) indicate the averaged DSL over all stations within each dry day frequency bin, the bin size is 0.02.

4. Discussion

As one of the important aspects of climate change, changes in dry spell duration could trigger the alteration of regional carbon and hydrological cycling (Field et al., 2012). According to the Clausius-Clapeyron (C-C) relationship, global warming could result in prolonged dry spell duration (Allen & Ingram, 2002; Trenberth et al., 2003, 2005). In this study, we examined the spatiotemporal trend of growing season DSL during 1976-2019 using the global in-suite precipitation observation data set. We found that global mean DSL during growing season significantly prolonged in 1976–1998 whereas no significant trend of that was observed in 1999–2019. The increase of growing season precipitation contributed to the diminished increasing trend of DSL during 1999–2019. The results implied that the evolution of dry spells did not entirely obey the C-C relationship but also was modulated by the inter-annual variation of precipitation. The impacts of precipitation changes induced by the alteration of atmospheric circulation under global warming should be taken into account in the estimation of DSL. The shift in trend of DSL from the increase during 1976–1998 to the decrease during 1999–2019 was mainly observed in stations that located in Europe, that could be attribute to the significant decrease of dry day frequency resulting from the increase of precipitation during 1999-2019. Using the daily rain gauge data, Zolina et al. (2010) reported that the wet spells duration have become longer in Europe. Although the increase of growing season precipitation was observed in subtropics in China during 1999-2019 comparing with the period of 1976–1998, the decrease trend of dry day frequency was not shown that resulted in no significant decrease trend of DSL during this period. Ma et al. (2015) found that the frequency of heavy precipitation (>25 mm day⁻¹) showed an increase trend after 2000, which indicated that the increased precipitation during 1999–2019 might be mainly from the heavy rain events and had little effects on the changes of DSL.

Changes in DSL can impact the terrestrial energy balance and ecosystem carbon and hydrological cycling. Ye and Fetzer (2019) examined the relationship between the dry spell duration and the increasing air temperature in summer during 1966–2010 basing on more than 500 stations in Russia, they found that hotter summers favor more frequent prolonged dry spells which could exacerbate drought and heat waves. Moreover, extended dry spells can enhance solar radiation which may result in the increase of evaporation and further induce the deficit of soil moisture. Increasing air temperature induced by prolonged DSL would result in the increase of atmospheric saturated water vapor capacity while the actual water vapor pressure would decrease resulting from the deficit soil moisture, that can result in the increase in vapor pressure deficit (VPD). In that case, the increased VPD would constraint the growth of terrestrial vegetation and further impact the ecosystem carbon and hydrological cycling (Yuan et al., 2019).



24711403, 2022, 4, Do

wiley.com/doi/10.1029/2022GH000611 by Deutsche Zentral Bibliothek Fuer Medizin

Medizin ische

Abt, Wiley Online Library on [26/01/2023]. See the Terms and Conditions (https://www.abt/10.1011/1011/10.1011/10.1011/10.1011/10.1011/1011/10.1011/1011/10.1011/101

on Wiley Online Library

for rule



Figure 5. Spatial distribution of the trends of the ratio of dry spell length (DSL) to growing season precipitation (DSL/Prec_GS) during (a) 1976–1998 and (b) 1999–2019. Global mean trends of the DSL/Prec_GS ratio during 1976–2019 (c). The small insets in (a and b) show the probability distribution functions of the DSL/ Prec_GS ratio trends during the two periods.

In this study, we found the trend of growing season DSL and precipitation shifted after the turning point year in 1998, the same turning point year was also reported by previous studies. Yuan et al. (2019) found that the global VPD showed a sharp increase trend after the late 1990s. The global mean surface air temperatures data indicated





Figure 6. Dry spell length trends in stations which experienced significant (P < 0.1) changes in growing season precipitation during the period of 1976–2019, 1976–1998, and 1999–2019, respectively. The small insets indicate the percentage of stations which showed negative relationship of DSL trends with growing season precipitation trends during the three periods.

a global warming hiatus between 1998 and 2012 (Fyfe et al., 2016; Kaufmann et al., 2011; Medhaug et al., 2017). The warming hiatus can slow the increase of atmospheric saturate vapor pressure, that would further ease the increase of residence time of atmospheric vapor. Wang et al. (2019) also reported that the global vegetation phenology was lack of widespread trends after 1998. The 1997–1998 EI Nino events was known to be one of the strongest in history, that affected the global precipitation changes (Kane, 1999). The linkages and interact mechanisms among the changes of DSL, increase of VPD as well as changes of growing season length should be explored in future studies.

5. Conclusion

Dry spell length plays an important role in regulating soil moisture dynamics, terrestrial energy exchange as well as vegetation growth. In this study, we examined the spatiotemporal trends of growing season DSL, basing on the global in-situ observation data set of daily precipitation from 1976 to 2019. The results showed the global mean growing season DSL significantly increased during 1976–1998 whereas no significant trend of that was observed during 1999–2019. The diminished increasing trend of DSL can attribute to the increase of growing season precipitation during 1999–2019. Our study indicated that the evolution of dry spells were modulated by the inter-annual variations of growing season precipitations. The linkages and interact mechanisms among the changes of DSL and precipitation characteristics should be explored in future studies.

Conflict of Interest

The authors declare no conflicts of interest relevant to this study.

Data Availability Statement

All data used in this study are publicly available, can be download from https://www.ncei.noaa.gov/data/global-summary-of-the-day/.

ational References

Allen, M. R., & Ingram, W. J. (2002). Constraints on future changes in climate and the hydrologic cycle. *Nature*, 419(6903), 228–232. https://doi.org/10.1038/nature01092

Bouagila, B., & Sushama, L. (2013). On the current and future dry spell characteristics over Africa. *Atmosphere*, 4(3), 272–298. https://doi. org/10.3390/atmos4030272

Caloiero, T., Caloiero, P., & Frustaci, F. (2018). Long-term precipitation trend analysis in Europe and in the Mediterranean basin. *Water and Environment Journal*, 32(3), 433–445. https://doi.org/10.1111/wej.12346

Abt, Wiley Online

on [26/01/2023]. See the

on Wiley Online Librai

24711403, 2022, 4, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2022GH000611 by Deutsche Zentral Bibliothek Fuer Medizin

Acknowledgments

This study was supported by the National Natural Science Foundation of China (Nos. 42101026, 31930072), Guangdong Basic and Applied Basic Research Foundation (2020A1515111145) and Innovation Group Project of Southern Marine Science and Engineering Guangdong Laboratory (Zhuhai) (No. 311021009).



- Chou, C., Chen, C.-A., Tan, P.-H., & Chen, K. T. (2012). Mechanisms for global warming impacts on precipitation frequency and intensity. *Journal of Climate*, 25(9), 3291–3306. https://doi.org/10.1175/jcli-d-11-00239.1
- Dore, M. H. (2005). Climate change and changes in global precipitation patterns: What do we know? *Environment International*, 31(8), 1167–1181. https://doi.org/10.1016/j.envint.2005.03.004
- Field, C. B., Barros, V., Stocker, T. F., & Dahe, Q. (2012). Managing the risks of extreme events and disasters to advance climate change adaptation: Special report of the intergovernmental panel on climate change. Cambridge University Press.
- Fischer, A. M., Keller, D. E., Liniger, M. A., Rajczak, J., Schär, C., & Appenzeller, C. (2015). Projected changes in precipitation intensity and frequency in Switzerland: A multi-model perspective. *International Journal of Climatology*, 35(11), 3204–3219. https://doi.org/10.1002/ joc.4162
- Fyfe, J. C., Meehl, G. A., England, M. H., Mann, M. E., Santer, B. D., Flato, G. M., et al. (2016). Making sense of the early-2000s warming slowdown. Nature Climate Change, 6(3), 224–228. https://doi.org/10.1038/nclimate2938
- Giorgi, F., Im, E.-S., Coppola, E., Diffenbaugh, N. S., Gao, X. J., Mariotti, L., & Shi, Y. (2011). Higher hydroclimatic intensity with global warming. *Journal of Climate*, 24(20), 5309–5324. https://doi.org/10.1175/2011jcli3979.1
- Groisman, P. Y., & Knight, R. W. (2008). Prolonged dry episodes over the conterminous United States: New tendencies emerging during the last 40 years. Journal of Climate, 21(9), 1850–1862. https://doi.org/10.1175/2007jcli2013.1
- Kane, R. P. (1999). Some characteristics and precipitation effects of the El Niño of 1997–1998. Journal of Atmospheric and Solar-Terrestrial Physics, 61(18), 1325–1346. https://doi.org/10.1016/S1364-6826(99)00087-5
- Kaufmann, R. K., Kauppi, H., Mann, M. L., & Stock, J. H. (2011). Reconciling anthropogenic climate change with observed temperature 1998– 2008. Proceedings of the National Academy of Sciences, 108(29), 11790–11793. https://doi.org/10.1073/pnas.1102467108
- Lambert, F. H., Stine, A. R., Krakauer, N. Y., & Chiang, J. C. H. (2008). How much will precipitation increase with global warming? *Eos, Transactions American Geophysical Union*, 89(21), 193–194. https://doi.org/10.1029/2008EO210001
- Lebedeva, M. G., Krymskaya, O. V., Lupo, A. R., Chendev, Y. G., Petin, A. N., & Solovyov, A. B. (2016). Trends in summer season climate for Eastern Europe and Southern Russia in the early 21st century. Advances in Meteorology, 2016, 5035086. https://doi.org/10.1155/2016/5035086
- Loaiciga, H. A., Valdes, J. B., Vogel, R., Garvey, J., & Schwarz, H. (1996). Global warming and the hydrologic cycle. Journal of Hydrology, 174(1), 83–127. https://doi.org/10.1016/0022-1694(95)02753-X
- Lu, H., Li, S., Ma, M., Bastrikov, V., Chen, X., Ciais, P., et al. (2021). Comparing machine learning-derived global estimates of soil respiration and its components with those from terrestrial ecosystem models. *Environmental Research Letters*, 16(5), 054048. https://doi. org/10.1088/1748-9326/abf526
- Ma, S., Zhou, T., Dai, A., & Han, Z. (2015). Observed changes in the distributions of daily precipitation frequency and amount over China from 1960 to 2013. Journal of Climate, 28(17), 6960–6978. https://doi.org/10.1175/jcli-d-15-0011.1
- Medhaug, I., Stolpe, M. B., Fischer, E. M., & Knutti, R. (2017). Reconciling controversies about the 'global warming hiatus'. Nature, 545(7652), 41–47. https://doi.org/10.1038/nature22315
- Sun, Y., Solomon, S., Dai, A., & Portmann, R. W. (2007). How often will it rain? Journal of Climate, 20(19), 4801–4818. https://doi.org/10.1175/jcli4263.1
- Sushama, L., Khaliq, N., & Laprise, R. (2010). Dry spell characteristics over Canada in a changing climate as simulated by the Canadian RCM. *Global and Planetary Change*, 74(1), 1–14. https://doi.org/10.1016/j.gloplacha.2010.07.004
- Toms, J. D., & Lesperance, M. L. (2003). Piecewise regression: A tool for identifying ecological thresholds. *Ecology*, 84(8), 2034–2041. https:// doi.org/10.1890/02-0472
- Trenberth, K. E., Dai, A., Rasmussen, R. M., & Parsons, D. B. (2003). The changing character of precipitation. Bulletin of the American Meteorological Society, 84(9), 1205–1218. https://doi.org/10.1175/bams-84-9-1205
- Trenberth, K. E., Fasullo, J., & Smith, L. (2005). Trends and variability in column-integrated atmospheric water vapor. Climate Dynamics, 24(7), 741–758. https://doi.org/10.1007/s00382-005-0017-4
- Wang, X., Xiao, J., Li, X., Cheng, G., Ma, M., Zhu, G., et al. (2019). No trends in spring and autumn phenology during the global warming hiatus. *Nature Communications*, 10(1), 2389. https://doi.org/10.1038/s41467-019-10235-8
- Wentz, F. J., Ricciardulli, L., Hilburn, K., & Mears, C. (2007). How much more rain will global warming bring? Science, 317(5835), 233–235. https://doi.org/10.1126/science.1140746
- Ye, H., & Fetzer, E. J. (2019). Asymmetrical shift toward longer dry spells associated with warming temperatures during Russian summers. Geophysical Research Letters, 46(20), 11455–11462. https://doi.org/10.1029/2019GL084748
- Ye, H., Fetzer, E. J., Behrangi, A., Wong, S., Lambrigtsen, B. H., Wang, C. Y., et al. (2016). Increasing daily precipitation intensity associated with warmer air temperatures over Northern Eurasia. *Journal of Climate*, 29(2), 623–636. https://doi.org/10.1175/jcli-d-14-00771.1
- Yuan, W., Zheng, Y., Piao, S., Ciais, P., Lombardozzi, D., Wang, Y., et al. (2019). Increased atmospheric vapor pressure deficit reduces global vegetation growth. *Science Advances*, 5(8), eaax1396. https://doi.org/10.1126/sciadv.aax1396
- Zolina, O., Simmer, C., Belyaev, K., Gulev, S. K., & Koltermann, P. (2013). Changes in the duration of European wet and dry spells during the last 60 years. *Journal of Climate*, 26(6), 2022–2047. https://doi.org/10.1175/jcli-d-11-00498.1
- Zolina, O., Simmer, C., Gulev, S. K., & Kollet, S. (2010). Changing structure of European precipitation: Longer wet periods leading to more abundant rainfalls. *Geophysical Research Letters*, 37(6). https://doi.org/10.1029/2010gl042468